Author's response to reviews

Title: Association between tobacco and alcohol use among young adult bar patrons: a cross-sectional study in three cities

Authors:

Nan Jiang (Nan.jiang@ucsf.edu)
Youn Ok Lee (younlee@rti.org)
Pamela M Ling (pling@medicine.ucsf.edu)

Version: 2 Date: 7 May 2014

Author's response to reviews:

Please see attached letter, which contains both cover letter and detailed response